

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Zaharieva, Anna

Working Paper Social welfare and wage inequality in search equilibrium with personal contacts

Working Papers, No. 459

Provided in Cooperation with: Center for Mathematical Economics (IMW), Bielefeld University

Suggested Citation: Zaharieva, Anna (2011) : Social welfare and wage inequality in search equilibrium with personal contacts, Working Papers, No. 459, Bielefeld University, Institute of Mathematical Economics (IMW), Bielefeld, https://nbn-resolving.de/urn:nbn:de:0070-pub-26717052

This Version is available at: https://hdl.handle.net/10419/81126

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Working Papers

Institute of Mathematical Economics

December 2011

Social Welfare and Wage Inequality in Search Equilibrium with Personal Contacts

459

Anna Zaharieva



IMW · Bielefeld University Postfach 100131 33501 Bielefeld · Germany



email: imw@wiwi.uni-bielefeld.de http://www.imw.uni-bielefeld.de/research/wp459.php ISSN: 0931-6558

Social Welfare and Wage Inequality in Search Equilibrium with Personal Contacts

Anna Zaharieva*

Institute of Mathematical Economics Bielefeld University, 33501 Bielefeld, Germany

December 8, 2011

Abstract

This paper incorporates job search through personal contacts into an equilibrium matching model with a segregated labour market. Job search in the public submarket is competitive which is in contrast with the bargaining nature of wages in the informal job market. Moreover, the social capital of unemployed workers is endogenous depending on the employment status of their contacts. This paper shows that the traditional Hosios (1990) condition continues to hold in an economy with family contacts but it fails to provide efficiency in an economy with weak ties. This inefficiency is explained by a network externality: weak ties yield higher wages in the informal submarket than family contacts. Furthermore, the spillovers between the two submarkets imply that wage premiums associated with personal contacts lead to higher wages paid to unemployed workers with low social capital but the probability to find a job for those workers is below the optimal level.

JEL classification: J23, J31, J64, D10

Keywords:

Personal contacts, family job search, social capital, wages, equilibrium efficiency

^{*}E-mail: azaharieva@wiwi.uni-bielefeld.de tel.: +49-521-106-5637, fax: +49-521-106-89005. I would like to thank Herbert Dawid for his comments and insights as well as seminar and session participants at the University of Bielefeld, University Paris 1 Pantheon-Sorbonne and the 2011 meeting of the Italian Association of Labour Economics.

1 Introduction

The aim of this article is to develop a labour market matching model with personal contacts and to investigate the implications of network effects for the equilibrium welfare and wage inequality. The seminal approach to address the question of equilibrium efficiency in modern labour economics is laid down in the contributions by Diamond, Mortensen and Pissarides¹. In a standard search and matching framework Hosios (1990) and later Pissarides (2000) explain the fundamentals of congestion externalities and prove existence of a unique value of the bargaining power parameter delivering efficiency to the decentralized equilibrium. Congestion externalities are internalized at the optimal value of the bargaining power. However an important limitation of this framework is an atomistic structure of the society where the possibility of information exchange in a group of connected workers is largely ignored.

Economic consequences of personal contacts and social networks are analyzed in a different strand of literature dating back to the original papers by Granovetter (1973, 1995) and Montgomery (1991, 1992, 1994). Montgomery (1994) considers a continuum of workers grouped in pairs in a Markov model of employment transitions. He demonstrates that an increase in weak-tie interactions reduces inequality in the employment rates and has a positive effect on the equilibrium welfare if inbreeding by employment status among weak ties is sufficiently low. Nevertheless the model is set in a partial equilibrium framework so that the effects of personal contacts on job creation and recruitment strategies by firms are not taken into account.

This paper combines the two strands of literature in a natural way by embedding the social structure of Montgomery (1994) into the traditional labour market model with search frictions. This allows to consider the implications of personal contacts for wage inequality and social welfare in a unified general equilibrium framework with endogenous wages and job-finding rates. The central issues addressed in this study are then the interaction between a search and a network externality and the channels of spillovers between the public and the informal job market. Wages in the public job market are set competitively, exploiting the fact that a more generous wage offer attracts a larger number of applications. The concept of competitive search employed in this paper is originally introduced in Moen (1997). In contrast, vacancy information in the informal job market is exclusively transmitted through employed personal contacts so that wages are set ex-post via the mechanism of Nash bargaining.

¹Diamond (1982), Mortensen (1982), Pissarides (1984, 1985) and Mortensen and Pissarides (1994).

In order to simplify the model it is assumed that every worker in the labour market has exactly one social link, which can be interpreted as a close relative, a friend or an acquaintance, so the economy is populated by an exogenous number of two-person groups (dyads). In the benchmark model of the paper a pair of connected individuals are fully sharing their labour income and therefore are treated as a single family or a household (strong ties). The model is further extended to relax the assumption of income sharing, which allows to analyze the inherent difference of a personal contact being a strong or a weak tie.

From the perspective of labour demand there is a free-entry of firms both into the public and the informal job market. Upon the decision to enter the labour market firms face a trade-off between a high cost vacancy in the public job market with a large number of searching unemployed workers versus a low cost vacancy only available to workers with an employed personal contact. The closest study to analyze social welfare in an equilibrium search model with a free-entry of firms is Cahuc and Fontaine (2009). The choice of search methods by firms is also endogenous in their model, however there is only one search method prevailing in the equilibrium, whereas in this study both search methods are simultaneously used by workers with employed social contacts.

The model predictions can be summarized in the following way. First of all, the model implies wage dispersion among equally productive risk-neutral workers. This is due to the ex-post differentiation of unemployed workers by social capital, which can be high or low, defined by the employment status of their contact. Only unemployed workers with high social capital have an additional access to the informal job market through their contacts, so the reservation wage of those workers is high. Wage competition between firms opening vacancies in the public job market combined with the endogenous differentiation of unemployed workers results in a segmentation of the public job market. Firms in low wage segment target at unemployed workers with low social capital and a low reservation wage, while the opposite is true for firms in a high wage segment.

Further, this paper considers the question of social welfare in an economy with personal contacts and shows that competitive search equilibrium with strong ties and bargaining in the informal job market is constraint efficient for the Hosios value of the bargaining power. The new contribution of this paper is to prove that wage dispersion between workers with high and low social capital in the public job market is maximized for the efficient value of the bargaining power. If the bargaining power parameter is low, meaning that wages paid in jobs obtained through personal contacts are low, then a higher value of this parameter has an enlarging effect on wage dispersion in the public job market. The functional relationship between the bargaining power and wage dispersion is reversed if the bargaining power parameter is large. This also means that both wage penalties and wage premiums in the informal job market lead to higher wages paid to unemployed workers with low social capital but the probability to find a job for those workers is below the optimal level.

The model is then extended to relax the assumption of income-sharing within a pair of connected workers. This allows to treat workers as friends or acquaintances helping each other to find a job, so the two economies with strong and weak ties can be compared. In the extended model workers bargaining over wages in the informal job market do not internalize the positive externality imposed on their social contacts inducing firms to pay higher wages. As a consequence competitive search equilibrium with weak ties and bargaining in the informal job market is not efficient at the Hosios value of bargaining power: too few job vacancies are filled in the informal job market. The implications of the described network externality for the public job market are twofold. At low values of the bargaining power the network externality has a neutralizing effect on the externality from search frictions. Workers with low social capital gain from a higher probability to find a job in the low wage segment of the public job market but their wages are lower. On the contrary, workers with high social capital face a lower job-finding rate but are compensated by higher wages. The overall effect on output is positive but these effects are reversed when the bargaining power parameter is above the efficient level.

Finally, theoretical predictions of the model are confronted with the empirical evidence. In general, the role of social networks and personal contacts is strongly emphasized in the empirical literature. The recent contributions are summarized in table 1 and show that between one and two-thirds of the employees in different countries have obtained their current job with a help of a friend or a relative². More specifically, the model predicts a positive correlation in the employment status of connected workers, both relatives and friends. On the empirical level the impact of family ties is closely investigated by Kramarz and Nordstrom Skans (2010). Their results show that a significant proportion of young employees in Sweden work for the same firm as their parents.

 $^{^{2}}$ An overview of the early empirical literature before 1990 is presented in Bewley (1999).

The effect of the employment status of friends is analyzed in Capellari and Tatsiramos (2010) who find that in the United Kingdom an additional employed friend increases the probability of finding a job by 3.7%.

Study	Incidence	Wage effects	Country	
Staiger (1990)	40%	Positive	United States	
Granovetter (1995)	56%	Positive	United States	
Pistaferri (1999)	47%	Negative	Italy	
Addison, Portugal (2002)	47%	Negative	Portugal	
Margolis, Simonnet (2003)	36%	Positive	France	
Delattre, Sabatier (2007)	34%	Negative	France	
Bentolila, Michelacci, Suarez (2008)	$31\% \\ 50\%$	Negative Negative	European Union United States	
Ponzo, Scoppa (2010)	31%	Negative	Italy	
Pelizzari (2010)	$38\%^*$	Positive Negative	Belgium, Netherlands Finland, Portugal, Italy, UK	

* – average for the European Union, 14 countries

Table 1: Empirical evidence on job search through personal contacts

The next theoretical prediction of the model concerns the effect of personal contacts on wages. Wages in the informal job market are set ex-post as a result of individual bargaining which is different from competitive wage setting in the public job market. Therefore personal contacts in the model can lead to penalties or premiums in wages depending on the parameter of bargaining power. Nevertheless the model predicts lower wages in the informal job market when personal contacts are strong rather than weak ties. Indeed, when bargaining workers account for the gain of a connected worker only if their labour income is shared, this has a weakening effect on their bargaining position and leads to lower wages. These predictions are similar to the empirical findings (see table 1), in particular, Pelizzari (2010) shows that in the European Union "... premiums and penalties to finding jobs through personal contacts are equally frequent and are of about the same size." (p. 1). However when the distinction between relatives and friends is explicit family ties tend to have a negative effect on wages (see Sylos Labini (2004), Delattre and Sabatier (2007), Kramarz and Nordstrom Skans (2010)). The plan of the paper is as follows. Section 2 contains an overview of the related literature while section 3 explains notation and the general economic environment of the model. Section 4 contains the labour market model with strong ties which is compared to the economy with weak ties in section 5. Section 6 contains welfare analysis of the decentralized equilibrium, whereas section 7 concludes the paper.

2 Related literature

Early economic studies on social contacts are Montgomery (1991, 1992, 1994) and Mortensen and Vishwanath (1994). The focus of Montgomery (1991) is on the effect of asymmetric information on wage inequality in the presence of the "inbreeding bias", implying clustering of workers with respect to their ability type. As a result the equilibrium is characterized by the positive correlation between ability and wages. Mortensen and Vishwanath (1994) consider the population of workers differing with respect to the probability of receiving job offers through personal contacts, they show that wages paid in jobs obtained through personal contacts are more likely to be higher than wage offers obtained through a direct application. This conclusion is questioned in the recent empirical literature, and moreover, "both the models of Montgomery (1991) and Mortensen and Vishwanath (1994) ignore what may be the most important role for network: to increase the job offer arrival rate." (p. 7, Margolis and Simonnet (2003)).

Recent theoretical literature on personal contacts is represented by the studies of Calvo-Armengol and Jackson (2004, 2007), Fontaine (2004, 2007, 2008) and Bentolila et al. (2010). A larger overview of this literature can be found in Ioannides and Datcher Loury (2004). Calvo-Armengol and Jackson (2004) examine a model of the transmission of job information through a network of social contacts and show that information passing leads to positive correlation between the employment status of agents who are directly or indirectly connected in the network. This effect is also present in the current study but in a richer equilibrium framework with endogenous wages and job-finding rates permitting analysis of the equilibrium welfare.

Calvo-Armengol and Jackson (2007) extend their initial result by showing that networks of agents that start with a worse wage status will have higher drop-out rates and persistently lower wages. The negative effect of social networks on wages is also demonstrated in Bentolila et al. (2010). In particular they show that social contacts can generate a mismatch between the occupational choice and the productive advantage of the worker leading to wage penalties in jobs obtained through personal contacts. The possibility of wage penalties is also included in the present study if the bargaining power of workers in the informal job market is low, however this wage effect is reversed if the bargaining power parameter is high. This framework is more general and allows to differentiate welfare implications of social networks in both types of labour markets with wage premiums and wage penalties.

Fontaine (2004) considers policies aiming at increasing individuals social capital by enlarging the access to networks and shows that such policies can increase the congestion externalities and induce firms to substitute employee referrals for job advertising. Eventually unemployment can increase and welfare decrease. The theoretical framework of this study is similar to the present work, but the results are different. The interaction between a search and a network externality is not discussed in Fontaine (2004) so the decentralized equilibrium is efficient under the Hosios value of the bargaining power. This is not the case in the present study where weak ties give rise to the inefficiency under the benchmark value of the bargaining power.

The paper is also related to the literature on search externalities and social welfare in an economy with heterogeneous agents. Gautier (2002) shows that mixing two types of workers (high and low skilled) in a single labour market generates additional pooling externalities. Blazquez and Jansen (2008) analyze welfare in this economy and prove that pooling compresses the wage distribution, therefore higher wages of low-ability workers discourage the creation of unskilled jobs. A straightforward extension of the present study to the case of random search and bargaining allows to conclude that the effect of pooling on job creation is reversed. Endogenous heterogeneity of workers with equal productivities leads to lower average wages in the public submarket and encourages firms to open jobs. This finding highlights the importance of the source of worker heterogeneity for the equilibrium efficiency.

Finally, this paper relates to the recent literature on family job search (joint search) represented by Guler et al. (2009) and Ek and Holmlund (2010). This literature shows that joint decisions by spouses give rise to different economic outcomes from the model of single agents. Nevertheless, the major focus of these studies is on income sharing within a family and the possibility to exchange job information between partners is not considered. Therefore this research study is the first to combine the literature on family job search with the literature on social networks. The combined approach shows that risk aversion is not a necessary condition to generate wage dispersion in a model of

family job search. Both this study and Fontaine (2008) show that information sharing between connected workers gives rise to endogenous wage dispersion even if workers are risk-neutral.

3 Labour market modeling framework

The labour market is characterized by the following properties. There is a unit mass of infinitely lived risk neutral workers and an endogenous number of firms, both workers and firms are ex-ante identical and discount the future at rate r. Every worker has exactly one social link, which can be interpreted as a close relative, a friend or an acquaintance. In the baseline model of the paper a pair of connected individuals is treated as a family with a full income-sharing within the household (strong ties). The model extension presented in section 5 considers consequences for the labour market once the income-sharing assumption is relaxed and pairs of connected workers are treated as friends or acquaintances helping each other to find a job (weak ties).

Every worker can be either unemployed, receiving the value of leisure z and searching for a job or employed and producing output y > z. Therefore all pairs of workers can be split into three mutually exhaustive groups: employed, mixed or unemployed. The total number of worker-pairs in each group is denoted p_e , p_m and p_u respectively:

$$p_e + p_m + p_u = 0.5$$

Every firm entering the labour market has an option to open a vacancy in the public job market with a high flow cost $c + \rho$ or in the informal job market with a low cost c. Vacancy information in the informal job market is transmitted through employed personal contacts, therefore only unemployed workers in mixed pairs have access to vacancies in the informal job market. In contrast every unemployed worker in the economy has access to vacancy information posted in the public job market. This creates a trade off for the firm: a costly public vacancy with a high number of searching workers $2p_u + p_m$ versus a low cost informal vacancy with a low number of searchers p_m . On-the-job search is prohibited, so that employed workers always forward job information to their unemployed contacts. This model structure implies that unemployed workers searching in the public job market are endogenously differentiated into two groups – with high or low social capital – depending on the employment status of a connected worker. The concept of competitive search, which was originally introduced in Moen (1997), is used to model search frictions in the public job market. Here firms post vacancies with exact information about the wage, while workers observe vacancy information and direct their search to particular jobs. It is assumed that firms commit to the posted employment contract. This wage-setting mechanism provides foundations for the wage competition between employers: firms offering higher wages are more likely to fill their open vacancies as opposed to the firms with low wage offers.

Endogenous heterogeneity of unemployed workers combined with competitive search implies that the public labour market is segmented into the submarket with low wages w_0 and short waiting queues, targeting at workers with low social capital, and a submarket with high wages w_1 and longer waiting queues, targeting at workers with high social capital. Let v_0 and v_1 denote the total number of vacancies in a low and high wage submarket respectively. Both unemployed workers and firms correctly anticipate the number of job matches m_i and the market tightness θ_i , in each of the submarkets i = 0, 1:

$$m_0 = m(2p_u, v_0)$$
 $\theta_0 = \frac{v_0}{2p_u}$ and $m_1 = m(p_m, v_1)$ $\theta_1 = \frac{v_1}{p_m}$

In contrast to the public job market, wages obtained through personal contacts (w_2) are not competitive, but set ex-post via the concept of Nash bargaining. Therefore search through personal contacts is random with a total number of job matches m_2 and the market tightness θ_2 given by:

$$m_2 = m(p_m, v_2) \qquad \theta_2 = \frac{v_2}{p_m}$$

The matching function m_i , i = 0, 1, 2 is assumed to be increasing in both arguments – unemployment and vacancies, concave, and exhibiting constant returns to scale. Then the job finding rate $\lambda(\theta_i)$ and the vacancy filling rate $q(\theta_i)$ are given by:

$$q(\theta_i) = \frac{m_i}{v_i} = q_0 \theta_i^{-\eta} \qquad \lambda(\theta_i) = \theta_i q(\theta_i) = q_0 \theta_i^{1-\eta}, \qquad i = 0, 1, 2$$

where $0 < \eta < 1$ is the elasticity of the job filling rate $q(\theta_i)$. Any job can be destroyed for exogenous reasons with a Poisson destruction rate δ . Upon a separation the worker becomes unemployed and the firm may open a new job.

4 Search equilibrium with personal contacts

4.1 Endogenous social capital

Let U and U_e denote asset values of unemployed workers with an unemployed and an employed partner respectively. In the following the concept of social capital is applied in order to distinguish the two types of unemployed workers³. The social capital is called high if the dyad partner of the worker is employed and transmits job information between the worker and the informal job market. The social capital is low if the dyad partner of the worker is unemployed. This means that the social capital is endogenous and is reflected in variables U and U_e .

Further, let W_u^i and W_e^i denote asset values of workers employed at wage w_i with an unemployed and an employed partner. Note that the subindex $\{u, e\}$ shows the employment status of a connected worker. Then, using the continuous time Bellman equations, asset values U, U_e, W_u^i and W_e^i can be written as:

$$rU = z + \lambda(\theta_0)(W_u^0 - U) + \lambda(\theta_0)(U_e - U)$$

$$(4.1)$$

$$rU_e = z + \lambda(\theta_1)(W_e^1 - U_e) + \lambda(\theta_2)(W_e^2 - U_e) - \delta(U_e - U)$$
(4.2)

$$rW_u^i = w_i - \delta(W_u^i - U) + (\lambda(\theta_1) + \lambda(\theta_2))(W_e^i - W_u^i), \qquad i = 0, 1, 2$$
(4.3)

$$rW_e^i = w_i - \delta(W_e^i - U_e) - \delta(W_e^i - W_u^i), \qquad i = 0, 1, 2$$
(4.4)

Labour market transitions for the special case $w_1 = w_2$ are illustrated in figure 1.

Consider an unemployed pair of workers, both partners are searching in the low wage segment of the public labour market with a job-finding rate $\lambda(\theta_0)$ and a wage w_0 . When either of the workers finds a job, the asset value of this worker is increased to the level W_u^0 with a corresponding job rent $R_u^0 \equiv W_u^0 - U$, while the surplus value of the connected worker is increased to U_e . The gain of the unemployed worker $\Delta U = U_e - U$ is twofold, on the one hand, the worker starts searching in a high wage segment of the public labour market with a high wage w_1 and the job-finding rate $\lambda(\theta_1)$, on the other, the worker obtains access to the informal job market through the employed personal contact. Value gain of the unemployed worker ΔU is then given by:

$$\Delta U = U_e - U = \frac{\lambda(\theta_1)R_e^1 + \lambda(\theta_2)R_e^2 - \lambda(\theta_0)R_u^0}{r + \delta + \lambda(\theta_0)}$$
(4.5)

where $R_e^1 = W_e^1 - U_e$, $R_e^2 = W_e^2 - U_e$ are, respectively, worker rents in the case of

 $^{^{3}}$ See Coleman (1988) for the definition of social capital.



Figure 1: Competitive search with personal contacts, $\bar{\lambda} = \lambda(\theta_1) + \lambda(\theta_2), w_2 = w_1$

accepting a job at wage w_1 in the public job market or a wage w_2 in the informal job market. However, not only unemployed workers gain from a better employment status of their partner. The gain of the employed worker in the event when the unemployed partner finds a job is denoted by $\Delta \Phi = W_e^i - W_u^i$, it results from the fact, that the partner will have a higher surplus value U_e rather than a low value U if the job is destroyed. Therefore the surplus gain $\Delta \Phi$ is given by:

$$\Delta \Phi = W_e^i - W_u^i = W_e^i - W_u^i = \frac{\delta \Delta U}{r + 2\delta + \lambda(\theta_1) + \lambda(\theta_2)} < \Delta U$$
(4.6)

Note that value gains of a connected worker ΔU and $\Delta \Phi$ are endogenous in the model.

4.2 Labour market with strong ties

Throughout the rest of section 4 consider an economy where dyad partners are fully sharing their income and therefore are treated as members of the same family and household. This is the case of strong social ties. Let P_u denote asset value of the unemployed household, so that $P_u = 2U$, similarly $P_m^j = U_e + W_u^j$ – asset value of the mixed household where one of the two family members is employed at wage w_j , j = 0, 1, 2. Finally let $P_e^{ij} = W_e^i + W_e^j$ denote surplus of the employed household earning wages w_i and w_j , i, j = 0, 1, 2. Then Bellman equations for P_u , P_m^j and P_e^{ij} are written as:

$$rP_u = 2z + 2\lambda(\theta_0)(P_m^0 - P_u)$$
(4.7)

$$rP_m^j = z + w_j + \lambda(\theta_1)(P_e^{1j} - P_m^j) + \lambda(\theta_2)(P_e^{2j} - P_m^j) - \delta(P_m^j - P_u)$$
(4.8)

$$rP_e^{ij} = w_i + w_j - \delta(P_e^{ij} - P_m^i) - \delta(P_e^{ij} - P_m^j)$$
(4.9)

The net job rent of the unemployed household if one of the workers finds a job $P_m^0 - P_u$ can be expressed as follows:

$$(r+\delta)(P_m^0 - P_u) = z + w_0 - rP_u + \lambda(\theta_1)(P_e^{10} - P_m^0) + \lambda(\theta_2)(P_e^{20} - P_m^0)$$
(4.10)

For a given vector of variables $\{w_1, \theta_1, w_2, \theta_2\}$ and therefore for fixed surplus values $P_e^{10} - P_m^0$ and $P_e^{20} - P_m^0$ equation (4.7) describes an indifference curve of the unemployed household searching in the low wage segment of the public labour market. The household is indifferent between obtaining a higher wage w_0 yielding a higher job rent $P_m^0 - P_u$ combined with a low job-finding rate $\lambda(\theta_0)$ versus a low wage w_0 combined with a high job-finding rate $\lambda(\theta_0)$. The slope of the indifference curve of the unemployed household $(P_u = cst)$ in the variable space $\{\theta_0, w_0\}$ is then obtained from:

$$\lambda'(\theta_0)\frac{d\theta_0}{dw_0}(P_m^0 - P_u) + \lambda(\theta_0)\frac{1}{r+\delta} = 0$$
(4.11)

This indifference curve is decreasing and convex in the space $\{\theta_0, w_0\}$. The total job rent $P_m^0 - P_u$ can be decomposed into the personal gain of the worker R_u^0 and the partner's gain ΔU : $P_m^0 - P_u = R_u^0 + \Delta U$, it can then be expressed as:

$$P_m^0 - P_u = \frac{w_0 - z + \lambda(\theta_1)(P_e^{10} - P_m^0) + \lambda(\theta_2)(P_e^{20} - P_m^0)}{r + \delta + 2\lambda(\theta_0)}$$
(4.12)

Further, the net job rent of the mixed household $P_e^{i0} - P_m^0$, when one of the members is employed at wage w_0 and the unemployed member finds a job at wage w_i , can be expressed as:

$$(r+2\delta)(P_e^{i0} - P_m^0) = w_i + w_0 - rP_m^0 + \delta \frac{w_i - w_0}{r+\delta}, \quad i = 1, 2$$
(4.13)

Note that surplus values $P^{10} - P_m^0$ and $P^{20} - P_m^0$ are independent of variables $\{w_0, \theta_0\}$ for a given value of P_u , which also means that these surplus values do not depend on the partner's wage:

$$P_e^{10} - P_m^0 = \frac{w_1 - z - \lambda(\theta_2)(P_e^{20} - P_m^0) + \delta(P_m^1 - P_u)}{r + 2\delta + \lambda(\theta_1)}$$
(4.14)

$$P_e^{20} - P_m^0 = \frac{w_2 - z - \lambda(\theta_1)(P_e^{10} - P_m^0) + \delta(P_m^2 - P_u)}{r + 2\delta + \lambda(\theta_2)}$$
(4.15)

Therefore all of the unemployed workers in mixed households search in the same high wage segment of the public labour market. This simplification of the model is attributed to the assumption of risk neutrality. The total gain of the household $P_e^{i0} - P_m^0$ can be similarly decomposed into the gain of the worker and the gain of the partner: $P_e^{i0} - P_m^0 = R_e^i + \Delta \Phi.$

For a given vector of variables $\{w_0, \theta_0, w_2, \theta_2\}$ the indifference curve of the mixed household where one worker is employed at wage w_0 is given by: $P_m^0 = cst$. Unemployed family members in a mixed household face a similar trade off between a high wage w_1 and therefore a high rent value $P_e^{10} - P_m^0$ combined with a low job arrival rate $\lambda(\theta_1)$ versus a low wage w_1 combined with a high job arrival rate $\lambda(\theta_1)$. The slope of the indifference curve $P_m^0 = cst$ in the space $\{\theta_1, w_1\}$ is then given by:

$$\lambda'(\theta_1) \frac{d\theta_1}{dw_1} (P_e^{10} - P_m^0) + \lambda(\theta_1) \frac{1}{r+\delta} = 0$$

This indifference curve is similarly decreasing and concave in the variable space $\{\theta_1, w_1\}$, however, it will be shown later that $P_e^{10} - P_m^0$ is smaller than $P_m^0 - P_u$ despite the fact that $w_1 > w_0$. Indeed given the equal productivity of workers, it should be the case that the rent gain of a household with a better outside option is lower than the gain of a household with a worse outside opportunity. This means that the indifference curve $P_u = cst$ is flatter than $P_m^j = cst$ in the space $\{\theta, w\}$.

4.3 Firms: wage determination

Firms are free to open a vacancy in the public labour market with a flow cost $c + \rho$ or in the informal market with a lower cost c. In addition, firms can freely choose between the two segments within the public labour market. Let V^0 and V^1 denote asset values of an open vacancy in a low/high wage segment of the public labour market, respectively, and V^2 – vacancy value in the informal job market. Bellman equations for V^0 , V^1 and V^2 are then given by:

$$rV^{i} = -(c+\rho) + q(\theta_{i})(J^{i} - V^{i}), \quad i = 0, 1$$
(4.16)

$$rV^2 = -c + q(\theta_2)(J^2 - V^2)$$
(4.17)

where J^0 , J^1 and J^2 are the corresponding asset values of a filled job:

$$rJ^{i} = y - w_{i} - \delta J^{i} \quad i = 0, 1, 2$$
(4.18)

Upon the decision to open a vacancy in the public job market firms face a similar trade-off as households. Paying a higher wage w_i , i = 0, 1 should be compensated by a higher probability to fill the job $q(\theta_i)$. It can be shown that the firm's indifference curves $V^i = cst$ are downward-sloping and convex in the space $\{\theta_i, w_i\}$. For given values $\{w_1, \theta_1, w_2, \theta_2\}$ denoted as information set I_0 firms in the low wage segment maximize their surplus V^0 , with respect to a combination $\{\theta_0, w_0\}$ and subject to the worker indifference curve $P_u = cst$:

$$V^{0}(P_{u}, I_{0}) = \max_{w_{0}, \theta_{0}} V^{0}(w_{0}, \theta_{0}) \qquad \text{s.t.} \qquad P_{u}(w_{0}, \theta_{0}, I_{0}) = cst$$
(4.19)

Solution of this maximization problem with a free-entry of firms meaning that in the equilibrium $V^0 = 0$ gives rise to the following rent-sharing condition:

$$J^{0} = \frac{(1-\eta)}{\eta} (P_{m}^{0} - P_{u}), \quad \text{where} \quad P_{m}^{0} - P_{u} = R_{u}^{0} + \Delta U$$
(4.20)

This equation is an extension of the result by Moen (1997) for the case of family job search. The wage w_0 is then given by:

$$w_0 = \eta y + (1 - \eta)[rU - (\lambda(\theta_1) + \lambda(\theta_2))\Delta\Phi - (r + \delta)\Delta U]$$
(4.21)

$$= \eta y + (1 - \eta)[rU - (r + 2\delta)(\Delta U - \Delta \Phi)]$$

$$(4.22)$$

There are two new terms in the reservation wage of the worker. The first of them, $(\lambda(\theta_1) + \lambda(\theta_2))\Delta\Phi$ is a future gain of the worker once the partner finds a job, while the second $(r + \delta)\Delta U$ is an immediate gain of the partner due to the possibility to search in the informal job market. Both gains act to reduce the reservation wage of the worker. Intuitively individuals are ready to work for lower wages if their partners and household members gain from additional job opportunities.

Similarly for given values $\{w_0, \theta_0, w_2, \theta_2\}$ denoted as information set I_1 firms in the high wage segment maximize their surplus V^1 with respect to a combination $\{w_1, \theta_1\}$ and subject to the worker indifference curve $P_m^0 = cst$:

$$V^{1}(P_{m}^{0}, I_{1}) = \max_{w_{1}, \theta_{1}} V^{1}(w_{1}, \theta_{1}), \qquad \text{s.t.} \qquad P_{m}^{0}(w_{1}, \theta_{1}, I_{1}) = cst$$
(4.23)

This maximization problem combined with a free-entry requirement $V^1 = 0$ gives rise

to the following rent-sharing condition:

$$J^{1} = \frac{(1-\eta)}{\eta} (P_{e}^{10} - P_{m}^{0}), \quad \text{where} \quad P_{e}^{10} - P_{m}^{0} = R_{e}^{1} + \Delta\Phi$$
(4.24)

and the following wage equation:

$$w_1 = \eta y + (1 - \eta)[rU_e + \delta\Delta\Phi - (r + \delta)\Delta\Phi]$$
(4.25)

$$= \eta y + (1 - \eta)[rU + r(\Delta U - \Delta \Phi)]$$
(4.26)

The first new term in the reservation wage $\delta \Delta \Phi$ is a future surplus loss of the worker once the partner loses the job, while the second term $(r+\delta)\Delta \Phi$ is an immediate gain of the partner. Here again the immediate surplus gain of the partner $\Delta \Phi$ is reducing the reservation wage of the worker. The partial equilibrium in the public job market is illustrated in figure 2. Firms are indifferent between the two segments since $V^0 = V^1 = 0$.



Figure 2: Segmentation in the public labour market

Comparison of equations (4.21) and (4.25) allows to evaluate the wage difference $w_1 - w_0$ showing the extent of wage dispersion in the public job market resulting from the introduction of personal contacts:

$$w_1 - w_0 = (1 - \eta)2(r + \delta)(\Delta U - \Delta \Phi)$$
 (4.27)

$$= (1-\eta)2(r+\delta)\Big[\frac{r+\delta+\lambda(\theta_1)+\lambda(\theta_2)}{r+2\delta+\lambda(\theta_1)+\lambda(\theta_2)}\Big]\Delta U$$
(4.28)

The more valuable is the access to the informal job market ΔU , the higher is the dif-

ference in the reservation wages of unemployed workers with different levels of social capital $2(r+\delta)(\Delta U - \Delta \Phi)$. By implication, a higher difference in the reservation wages leads to a higher wage dispersion in the public job market $w_1 - w_0$.

In the informal job market wages are determined ex-post, after the meeting between the firm and the unemployed worker. I use the concept of Nash bargaining in order to determine wage w_2 , the rent-sharing condition with $V^2 = 0$ and β denoting the worker's bargaining power is then:

$$J^{2} = \frac{(1-\beta)}{\beta} (P_{e}^{20} - P_{m}^{0}), \quad \text{where} \quad P_{e}^{20} - P_{m}^{0} = R_{e}^{2} + \Delta\Phi$$
(4.29)

with the following equation for wage w_2 :

$$w_2 = \beta y + (1 - \beta)[rU + r(\Delta U - \Delta \Phi)]$$
(4.30)

Clearly $w_2 = w_1$ if and only if $\beta = \eta$ and $w_2 > (<)w_1$ if and only if $\beta > (<)\eta$. Intuitively, job search through personal contacts is associated with wage premiums if $\beta > \eta$ and it leads to wage penalties otherwise. This completes the analysis of wages.

4.4 The decentralized equilibrium

The free entry of firms into every of the three submarkets implies that in the equilibrium $V^i = 0, i = 0, 1, 2$. Inserting these conditions into the asset value equations for V^i produces the following:

$$\frac{c+\rho}{q(\theta_i)} = J^i, \quad i = 0, 1 \qquad \frac{c}{q(\theta_2)} = J^2$$
 (4.31)

The left hand-side of these equations is the expected cost of opening a vacancy, since $q(\theta_i)$, i = 0, 1, 2, describes expected duration of the open vacancy. Expected cost of a vacancy in the equilibrium should be equal to the present value of flow profits from a filled job J^i . The rent-splitting equations (4.20), (4.24) and (4.29) imply that firms in the low wage segment of the public job market obtain fraction $(1 - \eta)$ of the total job surplus $S^0 \equiv J^0 + P_m^0 - P_u$. Firms in the high wage segment obtain a similar fraction of the total job surplus $S^1 \equiv J^1 + P_e^{10} - P_m^0$, while firms operating in the informal job market obtain a fraction $(1 - \beta)$ of the total surplus $S^2 = J^2 + P_e^{20} - P_m^0$, this means:

$$\frac{c+\rho}{q(\theta_i)} = (1-\eta)S^i, \quad i = 0, 1 \qquad \frac{c}{q(\theta_2)} = (1-\beta)S^2$$
(4.32)

A larger surplus value S^i attracts more entrants into the submarket, which is reflected in a higher value of the market tightness θ_i , i = 0, 1, 2. Equations (4.10) and (4.13) allow to rewrite surplus values S^0 , S^1 and S^2 as follows:

$$(r+\delta)S^{0} = y + z - rP_{u} + \lambda(\theta_{1})(P_{e}^{10} - P_{m}^{0}) + \lambda(\theta_{2})(P_{e}^{20} - P_{m}^{0})$$

(r+2\delta)S^{1} = y + w_{0} - rP_{m}^{0} + \delta J^{0} and S^{2} = S^{1}

Note that the total surplus S^2 does not directly depend on the exact surplus split between the firm and the worker and therefore does not directly depend on β , which means that $S^1 = S^2$. This equality allows to express the market tightness θ_2 in the equilibrium as a linear function of θ_1 :

$$q(\theta_2) = \frac{c(1-\eta)}{(c+\rho)(1-\beta)}q(\theta_1) \qquad \Rightarrow \qquad \theta_2 = \theta_2(\theta_1), \qquad \frac{\partial \theta_2(\theta_1)}{\partial \theta_1} > 0$$

Intuitively a larger surplus $S^1 = S^2$ has a positive effect on both variables θ_1 and θ_2 . Moreover it can be shown that the benchmark case $\beta = \eta$ implies that $\theta_2 > \theta_1$. If $w_1 = w_2$ then more firms exploit the cost advantage of the informal job market. Further, it can be shown that there exists a threshold value $\hat{\beta}$ such that:

$$\begin{cases} \text{if } \beta > \hat{\beta} \quad \text{then } \theta_2 < \theta_1 \\ \text{if } \beta < \hat{\beta} \quad \text{then } \theta_2 > \theta_1 \end{cases} \quad \text{where} \quad \hat{\beta} = \eta + \frac{\rho(1-\eta)}{c+\rho} \end{cases}$$

Using the functional relationship $\theta_2 = \theta_2(\theta_1)$ allows to simplify the characterization of the equilibrium to a vector of variables $\{\theta_0, \theta_1\}$. Using expressions $P_m^0 - P_u = \eta S^0$, $P_e^{20} - P_m^0 = \eta S^1$ and $P_e^{10} - P_m^0 = \beta S^1$ allows the following reformulation of surplus variables S^0 and S^1 :

$$S^{0} = \frac{y - z + [\eta \lambda(\theta_{1}) + \beta \lambda(\theta_{2})]S^{1}}{r + \delta + 2\lambda(\theta_{0})\eta}$$

$$(4.33)$$

$$S^{1} = \frac{y - z + \delta S^{0}}{r + 2\delta + \eta\lambda(\theta_{1}) + \beta\lambda(\theta_{2})}$$

$$(4.34)$$

The system of equations (4.33)-(4.34) describes spillovers between the submarkets. Consider a worker with an unemployed partner, a larger surplus gain S^1 created in the event when the unemployed partner finds a job (at rate $\lambda(\theta_1)$ or $\lambda(\theta_2)$) has a positive effect on the current surplus value of this worker and therefore on the total surplus value S^0 . Now consider a worker with an employed partner, a larger surplus loss S^0 in the event when the partner loses the job (at rate δ) has a negative effect on the reservation value of the household P_m^0 and therefore a positive effect on the current surplus value S^1 . Lemma 1 describes the effects of variables θ_0 and θ_1 on surplus values $S^0(\theta_0, \theta_1)$ and $S^1(\theta_0, \theta_1)$.

Lemma 1: Denote $\alpha(\theta_1) = \lambda(\theta_1) + \frac{\beta}{\eta}\lambda(\theta_2(\theta_1))$ – weighted job-finding rate for workers with high social capital, so that $\alpha'(\theta_1) > 0$ then total surplus values $S^0(\theta_0, \theta_1)$ and $S^1(\theta_0, \theta_1)$ can be expressed as follows:

$$S^{0} = \frac{(y-z)(r+2\delta+2\eta\alpha(\theta_{1}))}{(r+2\eta\lambda(\theta_{0}))(r+2\delta+\eta\alpha(\theta_{1}))+\delta(r+2\delta)}$$
$$S^{1} = \frac{(y-z)(r+2\delta+2\eta\lambda(\theta_{0}))}{(r+2\eta\lambda(\theta_{0}))(r+2\delta+\eta\alpha(\theta_{1}))+\delta(r+2\delta)}$$

Moreover, $S^0(\theta_0, \theta_1)$ is a decreasing function of θ_0 but an increasing function of θ_1 while $S^1(\theta_0, \theta_1)$ is decreasing in both arguments, formally:

$$\frac{\partial S^0(\theta_0, \theta_1)}{\partial \lambda(\theta_0)} < 0 \quad \frac{\partial S^0(\theta_0, \theta_1)}{\partial \alpha(\theta_1)} > 0 \quad \frac{\partial S^1(\theta_0, \theta_1)}{\partial \lambda(\theta_0)} < 0 \quad \frac{\partial S^1(\theta_0, \theta_1)}{\partial \alpha(\theta_1)} < 0$$

Proof: Differentiate surplus variable S^0 with respect to $\alpha(\theta_1)$:

$$\frac{\partial S^0(\theta_0, \theta_1)}{\partial \alpha(\theta_1)} = \frac{(y-z)\eta(r+2\delta)(r+2\delta+2\eta\lambda(\theta_0))}{[(r+2\eta\lambda(\theta_0))(r+2\delta+\eta\alpha(\theta_1))+\delta(r+2\delta)]^2} > 0$$

Differentiate surplus variable S^1 with respect to $\lambda(\theta_0)$:

$$\frac{\partial S^1(\theta_0, \theta_1)}{\partial \lambda(\theta_0)} = -\frac{(y-z)2\eta\delta(r+2\delta+2\eta\alpha(\theta_1))}{[(r+2\eta\lambda(\theta_0))(r+2\delta+\eta\alpha(\theta_1))+\delta(r+2\delta)]^2} < 0$$

Intuitively a larger job finding rate $\lambda(\theta_0)$ has a positive effect on the present value of the unemployed household P^u which has a direct negative effect on surplus $S^0 = J^0 + P_m^0 - P_u$. There is then a spillover into the high wage segment since a lower value of S^0 is reducing the surplus value S^1 . Second, a larger weighted job finding rate $\alpha(\theta_1)$ has a direct positive effect on S^0 but also a negative effect on $S^1 = J^1 + P_e^{10} - P_m^0$ due to a higher present value P_m^0 . Lemma 1 shows that the direct positive effect of $\alpha(\theta_1)$ on S^0 is dominating, however S^1 is lower despite a higher value of S^0 .

The main conclusion following from lemma 1 is that the free-entry condition in the low wage public market segment describes an increasing functional relationship between variables θ_0 and θ_1 . A higher probability to find a job for the partner θ_1 has a positive effect on the total job surplus S^0 , the fraction $1 - \eta$ of this surplus accrues to firms and therefore has a positive effect on the job creation θ_0 :

$$\frac{c+\rho}{q(\theta_0)} = (1-\eta)S^0(\theta_0,\theta_1) \Rightarrow \quad \theta_0 = \theta_0(\theta_1), \quad \frac{\partial\theta_0(\theta_1)}{\partial\theta_1} > 0 \qquad (JC_0)$$

In contrast the free-entry condition in the high wage public market segment describes a negative relationship between variables θ_0 and θ_1 – the higher the job finding rate $\lambda(\theta_0)$ which means the easier is it to find an initial job, the lower is the surplus of this job S^0 . This has a negative effect on the surplus S^1 and a lower job creation θ_1 :

$$\frac{c+\rho}{q(\theta_1)} = (1-\eta)S^1(\theta_0,\theta_1) \Rightarrow \quad \theta_1 = \theta_1(\theta_0), \quad \frac{\partial\theta_1(\theta_0)}{\partial\theta_0} < 0 \qquad (JC_1)$$

The unique intersection between the increasing curve $\theta_0(\theta_1)$ and the decreasing curve $\theta_1(\theta_0)$ allows to obtain the equilibrium values of θ_0 and θ_1 , this is illustrated in figure 3. The equilibrium is defined in the following way:

Definition 1 A competitive search equilibrium with strong ties and bargaining in the informal job market is a vector of variables $\{P_u, P_m^j, P_e^{ij}, V^i, J^i, w_i, \theta_i\}, i, j = 0, 1, 2$ satisfying the asset value equations for workers (4.7), (4.8), (4.9), for firms (4.16), (4.17), (4.18), the three rent-sharing equations (4.20), (4.24), (4.29) and the free-entry conditions $V^i = 0$.



Figure 3: Equilibrium values of θ_0 and θ_1

Proposition 1 shows that there exists a unique search equilibrium with strong ties.

Proposition 1: There exists a unique competitive search equilibrium with strong ties and bargaining in the informal job market described in definition 1. The equilibrium market tightness variables $\{\theta_0, \theta_1\}$ are obtained from the following system of equations:

(a.) The job creation curve in the public job market (JC_0) describes a positive rela-

tionship between the market tightness variables θ_0 and θ_1 , specifically:

$$\frac{c+\rho}{q(\theta_0)} = \frac{(1-\eta)(y-z)(r+2\delta+2\eta\alpha(\theta_1))}{(r+2\eta\lambda(\theta_0))(r+2\delta+\eta\alpha(\theta_1))+\delta(r+2\delta)}$$

(b.) The job creation curve in the public job market (JC_1) describes a negative relationship between the market tightness variables θ_0 and θ_1 , specifically:

$$\frac{c+\rho}{q(\theta_1)} = \frac{(1-\eta)(y-z)(r+2\delta+2\eta\lambda(\theta_0))}{(r+2\eta\lambda(\theta_0))(r+2\delta+\eta\alpha(\theta_1))+\delta(r+2\delta)}$$

(c.) wage dispersion in the public job market Δw is given by:

$$\Delta w = \frac{2(1-\eta)\eta(r+\delta)(y-z)(\alpha(\theta_1)-\lambda(\theta_0))}{(r+2\eta\lambda(\theta_0))(r+2\delta+\eta\alpha(\theta_1))+\delta(r+2\delta)}$$

Wage dispersion $\Delta w = w_1 - w_0$ is increasing in θ_1 and decreasing in θ_0 .

Proof: Parts (a) and (b) follow directly from lemma 1. For part (c) differentiate $\Delta \tilde{w} = \Delta w/(2(1-\eta)\eta(r+\delta)(y-z))$ with respect to θ_1 :

$$\frac{\partial \Delta \tilde{w}}{\partial \theta_1} = \frac{(r+2\eta\lambda(\theta_0))(r+2\delta+\eta\lambda(\theta_0))+\delta(r+2\delta)}{[(r+2\eta\lambda(\theta_0))(r+2\delta+\eta\alpha(\theta_1))+\delta(r+2\delta)]^2} > 0$$

Corollary 1: Competitive equilibrium in the family search model with bargaining in the informal job market entails positive wage dispersion among equally productive risk-neutral workers. In particular:

$$\begin{cases}
w_{2} < w_{0} = w_{1} & \text{if } \beta = 0 \\
w_{0} < w_{1} & w_{2} < w_{1} & \text{if } 0 < \beta < \eta \\
w_{0} < w_{1} = w_{2} & \text{if } \beta = \eta \\
w_{0} < w_{1} < w_{2} & \text{if } \eta < \beta < 1 \\
w_{0} = w_{1} < w_{2} & \text{if } \beta = 1
\end{cases}$$
(4.35)

Corollary 1 shows that interior values of the bargaining power parameter $0 < \beta < 1$ lead to a segmented public labour market, so that $w_0 < w_1$. However, this segmentation disappears at the corner values of the bargaining power $\beta = 0$ and $\beta = 1$. This is due to the fact that the option to search in the informal job market $\lambda(\theta_2)(P_e^{2j} - P_m^j)$ has zero value at $\beta = 0$ and $\beta = 1$. If the bargaining power of workers is zero, then the total household surplus $P_e^{2j} - P_m^j$ is zero, whereas if β is converging to 1, too few firms will use the informal job market to post vacancies, so that θ_2 is converging to zero. Moreover, Ek and Holmlund (2010) prove that risk aversion is necessary to generate endogenous wage dispersion among equally productive workers in a model with family job search. Corollary 1 extends this result by showing that information sharing between household members can generate wage dispersion even if workers are risk neutral.

4.5 The equilibrium unemployment

Let p_u , p_m and p_e denote the number of unemployed, mixed and employed households respectively. The equilibrium values of these variables can be obtained from the following system of differential equations:

$$\begin{cases} \dot{p_u} = \delta p_m - 2\lambda(\theta_0)p_u \\ \dot{p_e} = (\lambda(\theta_1) + \lambda(\theta_2))p_m - 2\delta p_e \\ 0.5 = p_u + p_m + p_e \end{cases}$$
(4.36)

In the stationary equilibrium the inflow of households into a particular state should be equal to the outflow of households from this state, namely $\dot{p_u} = 0$, $\dot{p_m} = 0$, $\dot{p_e} = 0$. The number of households of each type is then:

$$p_u = \frac{0.5\delta^2}{[\delta^2 + \lambda(\theta_0)(2\delta + \lambda(\theta_1) + \lambda(\theta_2))]}$$
$$p_m = \frac{\delta\lambda(\theta_0)}{[\delta^2 + \lambda(\theta_0)(2\delta + \lambda(\theta_1) + \lambda(\theta_2)]}$$

and $p_e = 0.5 - p_u - p_m$. The number of unemployed households is falling in any of the job finding rates $\lambda(\theta_i)$, i = 0, 1, 2, in contrast, the number of employed households is increasing. The effects on the number of mixed households are inversely directed: p_m is falling in $\lambda(\theta_1)$ and $\lambda(\theta_2)$ but it is increasing in $\lambda(\theta_0)$.

Further consider the special case $\beta = \eta$, so that the wage distribution in the equilibrium is binary $w_1 = w_2$. Lemma 2 shows the distribution of workers by income categories $\{z, w_0, w_1\}$:

Lemma 2: Let $\beta = \eta$, then the equilibrium unemployment rate u and the fraction of workers employed at wage w_1 denoted f are given by:

$$u = \frac{\delta(\delta + \lambda(\theta_0))}{[\delta^2 + \lambda(\theta_0)(2\delta + \lambda(\theta_1) + \lambda(\theta_2))]} \qquad f = \frac{\lambda(\theta_1) + \lambda(\theta_2)}{\delta + \lambda(\theta_1) + \lambda(\theta_2)}$$

The fraction 1 - f of workers are employed at wage w_0 .

Proof: The equilibrium unemployment is given by $u = 2p_u + p_m$, while the total number of workers e_0 , e_1 employed at wages w_0 , w_1 are given by:

$$e_0 = \frac{2\delta(p_m + p_e)}{2\delta + \lambda(\theta_1) + \lambda(\theta_2)} \qquad e_1 = \frac{(\lambda(\theta_1) + \lambda(\theta_2))(p_m + p_e)}{2\delta + \lambda(\theta_1) + \lambda(\theta_2)} + p_e$$

The fraction f is then $e_1/(e_0 + e_1)$.

Note that the equilibrium unemployment is falling in all of the job finding rates $\lambda(\theta_i), i = 0, 1, 2$, but the wage distribution $\{1 - f, f\}$ is independent of the job finding rate $\lambda(\theta_0)$.

In addition, for the special case $\beta = \eta$, it can be shown that the conditional probability of being unemployed for a worker with an employed contact $P\{u|e\}$ is lower than the conditional probability of being unemployed with an unemployed contact $P\{u|u\}$:

$$P\{u|e\} = \frac{0.5p_m}{p_e + 0.5p_m} = \frac{\delta}{\delta + \lambda(\theta_1) + \lambda(\theta_2)} < P\{u|u\} = \frac{\delta}{\delta + \lambda(\theta_0)}$$

so the labour market exhibits a positive correlation in the employment status of workers within one family. This result follows from $\Delta w > 0$ and so $\lambda(\theta_1) + \lambda(\theta_2) > \lambda(\theta_0)$.

4.6 Comparative statics

Empirical studies presented in section 1 show that wages in jobs obtained through personal contacts can be higher or lower than wages obtained through a direct job application. This section addresses the effect of the bargaining power β , and therefore the effect of wage w_2 , on market tightness variables, wages and wage dispersion in the public job market. Clearly a larger bargaining power parameter β has a positive direct effect on wage w_2 and a negative effect on θ_2 since a larger wage in the informal job market reduces the number of open vacancies. The spillovers of this effect into the public job market are summarized in proposition 2:

Proposition 2: The economic effects of a larger bargaining power $0 < \beta < 1$ in the informal submarket on variables $\{\theta_0, \theta_1, \theta_2, w_0, w_1, w_2\}$ are summarized in table 2. **Proof:** Appendix I.

The effect of a higher bargaining power β is additionally illustrated on figure 4. For the corner case $\beta = 0$ jobs obtained through personal contacts pay exactly the

$\uparrow \beta$	θ_0	θ_1	θ_2	w_0	w_1	w_2
$\beta < \eta$	+	—	—	—	+	+
$\beta > \eta$	_	+	_	+	_	+

Table 2: Economic effects of a higher bargaining power

reservation wage of the worker and for this reason do not add any additional value, so that $w_0 = w_1$. The situation is similar for $\beta = 1$ meaning that $\theta_2 = 0$. For $0 < \beta < \eta$, the term $\beta\lambda(\theta_2)$ is falling in β due to a lower value of θ_2 that has a negative effect on $\alpha(\theta_1) = \lambda(\theta_1) + \frac{\beta}{\eta}\lambda(\theta_2)$ – the weighted job finding rate. This is raising the total job surplus S^0 and the market tightness θ_0 (see lemma 1). In contrast a lower value of S^1 implies a lower job creation θ_1 . There is then a reverse prediction for wages w_0 and w_1 . The situation is exactly the opposite for $\eta < \beta < 1$ when the effect of a higher bargaining power is dominating so the term $\beta\lambda(\theta_2)$ is increasing. The above analysis shows that wage dispersion in the public job market Δw achieves maximum at $\beta = \eta$ when the economic effect of the informal job market is maximized.



Figure 4: Economic effects of a higher bargaining power parameter β

On the intuitive level figure 4 shows that both wage premiums and wage penalties associated with personal contacts lead to higher wages paid to unemployed workers with low social capital but the probability to find a job for those workers is reduced.

5 Competitive search equilibrium with weak ties

This section investigates the role of income-sharing and the implications of this assumption for the equilibrium outcomes. If the assumption of full income-sharing is relaxed, a pair of connected workers can then be seen as friends or acquaintances (weak ties). Consider the bargaining problem between a firm and a worker in the informal job market. The total job surplus of the worker is given by:

$$(r+\delta)R_e^2 = w_2 - rU_e - \delta\Delta\Phi$$

Nash bargaining implies that the joint surplus $R_e^2 + J^2$ is shared in the proportion β , so that $\beta J^2 = (1 - \beta)R_e^2$. This gives rise to the following wage equation:

$$w_2 = \beta y + (1 - \beta)[rU + r(\Delta U - \Delta \Phi) + (r + \delta)\Delta \Phi]$$

Compare this to equation (4.30). Wage w_2 is higher *ceteris paribus* in an economy with weak ties due to the additional term $(r + \delta)\Delta\Phi > 0$. When the flow income is shared within a pair of connected workers the total job surplus is $R_e^2 + J^2 + \Delta\Phi$, so the worker obtains a share β of the total surplus net of the gain of a connected worker, specifically $R_e^2 = \beta(R_e^2 + J^2 + \Delta\Phi) - \Delta\Phi = \beta(R_e^2 + J^2) - (1 - \beta)\Delta\Phi$. Clearly this surplus is lower than $\beta(R^2 + J^2)$ in an economy with weak ties, so that workers in the informal job market demand higher wages when their dyad partner's gain is not taken into account.

In a public job market firms in a low wage segment maximize their surplus $V^0(w_0, \theta_0)$ subject to the worker indifference curve U = cst. The slope of this indifference curve is identical to the slope of 2U = cst which is an indifference curve of a connected pair of unemployed workers. This means that there are no changes in the equation for w_0 , and the total job surplus is split according to $\eta J^0 = (1 - \eta)(R_u^0 + \Delta U)$.

In a high wage segment of the public job market firms maximize their surplus $V^1(w_1, \theta_1)$ subject to the worker indifference curve $U_e = cst$:

$$V^{1}(U_{e}, I_{1}) = \max_{w_{1}, \theta_{1}} V^{1}(w_{1}, \theta_{1})$$
 s.t. $U_{e}(w_{1}, \theta_{1}, I_{1}) = cst$

where
$$(r+\delta)U_e = z + \lambda(\theta_1)R_e^1 + \lambda(\theta_2)R_e^2 + \delta U$$

The worker indifference curve $U_e = cst$ is steeper in the space $\{\theta, w\}$ than the household indifference curve $U_e + W_u^i = cst$. This gives rise to lemma 3:

Lemma 3: The slope of the worker indifference curve $U_e(w_1, \theta_1, I_1) = cst$, where I_1 denotes an information set $\{w_0, \theta_0, w_2, \theta_2\}$, is given by:

$$\lambda'(\theta_1)\frac{d\theta_1}{dw_1}R_e^1 + \frac{\lambda(\theta_1)}{r+\delta} + k\lambda'(\theta_1)\frac{d\theta_1}{dw_1}\Delta\Phi = 0$$
(5.37)

where
$$\frac{1}{k} = 1 + \frac{r}{\delta} \left[\frac{(r+2\delta)(r+\lambda(\theta_0)+\delta)}{(\lambda(\theta_1)+\lambda(\theta_2))(r+2\lambda(\theta_0))+\lambda(\theta_0)(r+2\delta))} \right] > 1,$$

This gives rise to the following equilibrium equation for w_1 :

$$w_1 = y\eta + (1 - \eta)[rU + r(\Delta U - \Delta \Phi) + (r + \delta)\Delta \Phi(1 - k)]$$

Proof: Appendix II.

Consider an increase in w_1 along the worker indifference curve $U_e = cst$. In the absence of income-sharing, the direct effect of changes in w_1 on the surplus of a connected worker W_u^i is not taken into account, whereas it is accounted for along the household indifference curve $U_e + W_u^i = cst$. The corresponding drop in the surplus of a connected worker is attributed to a lower flow probability for the partner to find a job $\lambda(\theta_1)$:

$$(r+\delta)W_u^i = w_i + \delta U + (\lambda(\theta_1) + \lambda(\theta_2))\Delta\Phi$$

so that

$$(r+\delta)\frac{\partial W_u^i}{\partial w_1} = \lambda'(\theta_1)\frac{\partial \theta_1}{\partial w_1}\Delta\Phi + (\lambda(\theta_1) + \lambda(\theta_2))\frac{\partial \Delta\Phi}{\partial w_1}$$

since U is constant along the household indifference curve $U_e + W_u^i = cst$:

$$U = \frac{\lambda(\theta_0)}{r + 2\lambda(\theta_0)} (U_e + W_u^0)$$

In contrast, the asset value of an unemployed worker U is decreasing along the worker indifference curve $U_e = cst$:

$$\Big[r + \frac{(r + \lambda(\theta_0))(r + \delta)}{\lambda(\theta_0)}\Big]\frac{\partial U}{\partial w_1} = \lambda'(\theta_1)\frac{\partial \theta_1}{\partial w_1}\Delta\Phi + (\lambda(\theta_1) + \lambda(\theta_2))\frac{\partial\Delta\Phi}{\partial w_1}$$

For $r \to 0$, expression in the square brackets is converging to δ , so that the indirect effect of a higher w_1 on U and therefore on U_e is converging to the direct effect of w_1 on W_u^i . The difference between the stronger direct and the weaker indirect effect is reflected in the auxiliary variable k, such that $\lim_{r\to 0} k = 1$. Therefore the wage w_1 is *ceteris paribus* higher than in the economy with household job search due to the additional term $(r + \delta)\Delta\Phi(1 - k)$ but this difference becomes negligibly small for $r \to 0$. In the following in order to focus on the network externality stemming from the informal job market consider the limiting case $r \to 0$. In the absence of income-sharing the surplus splitting rule in the informal job market implies $J^2 = (1 - \beta)(J^2 + R_e^2) = (1 - \beta)(S^2 - \Delta\Phi)$, where $S^2 = J^2 + R_e^2 + \Delta\Phi$ is the total job surplus. Denote $\tilde{\beta}$ – the implied bargaining power, such that $J^2 = (1 - \tilde{\beta})S^2$. The implied bargaining power $\tilde{\beta}$ is a fraction of the firm surplus J^2 in the total job surplus S^2 and can be obtained from:

$$\tilde{\beta} = \beta + \phi(\theta_0, \theta_1, \theta_2)(1 - \beta), \quad where \quad \phi(\theta_0, \theta_1, \theta_2) = \frac{\Delta\Phi}{S^2}$$
(5.38)

Variable $\phi(\theta_0, \theta_1, \theta_2)$ is a fraction of the gain of a connected worker $\Delta \Phi$ in the total job surplus S^2 . The implied bargaining power $\tilde{\beta}$ is larger then β for $0 < \beta < 1$ and captures changes in the informal job market stemming from a relaxed assumption of income sharing. If $\beta = 0$ wage dispersion in the public job market Δw is zero implying $\Delta \Phi = 0$ and $\tilde{\beta} = \beta = 0$. Similarly if $\beta = 1$ it is true that $w_2 = y$, so that differences in the reservation wages of workers are not reflected in wage w_2 . This means that $\tilde{\beta} = \beta = 1$.

Proposition 3 shows that the labour market equilibrium with weak social ties (for $r \to 0$) is a special case of the equilibrium with strong social ties where the total job surplus S^2 is split in the proportion $\tilde{\beta}$ between the worker and the firm:

Proposition 3: Consider the case $r \to 0$, competitive search equilibrium without income sharing is a special case of the search equilibrium described in definition 1 with the implied bargaining power parameter $\tilde{\beta}$:

$$\tilde{\beta} = \beta + (1 - \beta) \frac{\delta(\eta \lambda(\theta_1) + \beta \lambda(\theta_2) - \eta \lambda(\theta_0))}{2(\delta + \lambda(\theta_0)\eta)(\delta + \lambda(\theta_1) + \lambda(\theta_2)) - (1 - \beta)\delta\lambda(\theta_2)}$$
(5.39)

The equilibrium market tightness variables $\{\theta_0, \theta_1, \theta_2\}$ are obtained from the following system of equations:

(a.) The job creation condition in the low wage segment of the public job market:

$$\frac{c+\rho}{q(\theta_0)} = \frac{(1-\eta)(y-z)(\delta+\eta\lambda(\theta_1)+\beta\lambda(\theta_2))}{\eta\lambda(\theta_0)(2\delta+\eta\lambda(\theta_1)+\tilde{\beta}\lambda(\theta_2))+\delta^2}$$

(b.) The job creation condition in the high wage segment of the public job market:

$$\frac{c+\rho}{q(\theta_1)} = \frac{(1-\eta)(y-z)(\delta+\eta\lambda(\theta_0))}{\eta\lambda(\theta_0)(2\delta+\eta\lambda(\theta_1)+\tilde{\beta}\lambda(\theta_2))+\delta^2}$$

(c.) The job creation condition in the informal job market:

$$q(\theta_2) = \frac{c(1-\eta)}{(c+\rho)(1-\tilde{\beta})}q(\theta_1)$$

Proof: Appendix II.

Proposition 3 proves that relaxing the income-sharing assumption is equivalent to an exogenous shift of the bargaining power in the informal job market from β to $\tilde{\beta}$. Intuitively, the gain of a connected worker $\Delta \Phi$ is ignored in the bargaining process, this strengthens the bargaining position of workers and leads to higher wages w_2 in the informal job market. The implications of the income sharing assumption for the equilibrium efficiency are considered in the next section.

6 Social optimum: welfare maximization

Hosios (1990) and further Pissarides (2000) show, that the Nash wage equation is not likely to internalize search externalities resulting from the dependence of transition probabilities $\lambda(\theta_i)$ and $q(\theta_i)$ on the tightness of the market. Nevertheless Hosios (1990) proves that search externalities may be internalized, if $\beta = \eta$, where η is the elasticity of the job-filling rate $q(\theta_i)$. Further Moen (1997) strengthens this finding by showing that competitive search equilibrium gives rise to a worker surplus share of η , so the equilibrium in his model is constrained efficient.

This section investigates efficiency properties of the competitive search equilibrium with bargaining in the informal job market, and shows that the classical Hosios efficiency condition continues to hold in an economy with strong ties but it fails to deliver efficiency in the economy with weak social ties. To obtain this result, consider the problem of a social planner, whose objective is to maximize the present discounted value of output minus the costs of job creation:

$$\max_{\theta_{0},\theta_{1},\theta_{2}} \int_{0}^{\infty} e^{-rt} \left[p_{u}2z + p_{m}(z+y) + p_{e}2y - (c+\rho)\theta_{0}2p_{u} - (c+\rho)\theta_{1}p_{m} - c\theta_{2}p_{m} \right] dt$$
(6.40)

In addition, the social planner is subject to the same matching constraints as firms and workers, therefore the dynamics of unemployment is described by the system of differential equations (4.36). Proposition 4 presents solution of the planner optimization problem (6.40) subject to the equilibrium conditions \dot{p}_u and \dot{p}_e . **Proposition 4:** Consider a social planner choosing the optimal number of vacancies $v_0 = 2p_u\theta_0$, $v_1 = p_m\theta_1$ – in the public job market and $v_2 = p_m\theta_2$ – in the informal job market. Then the optimal job creation is:

$$\frac{c+\rho}{q(\theta_0)} = (1-\eta)\mu_u \qquad \frac{c+\rho}{q(\theta_1)} = (1-\eta)\mu_e \qquad \frac{c}{q(\theta_2)} = (1-\eta)\mu_e$$

where variables μ_u and μ_e are obtained from the following system of equations:

$$\mu_{u} = \frac{y - z + (c + \rho)[2\theta_{0} - \theta_{1}] - c\theta_{2} + (\lambda(\theta_{1}) + \lambda(\theta_{2}))\mu_{e}}{r + \delta + 2\lambda(\theta_{0})}$$
(6.41)

$$\mu_e = \frac{y - z + (c + \rho)\theta_1 + c\theta_2 + \delta\mu_u}{r + 2\delta + \lambda(\theta_1) + \lambda(\theta_2)}$$
(6.42)

Proof: Appendix III.

Costate variables μ_u and μ_e in the dynamic optimization problem of the social planner correspond to the surplus values S^0 and $S^1 = S^2$ in the decentralized equilibrium. To see this insert the job creation conditions of the social planner $(c + \rho)\theta_0 =$ $\lambda(\theta_0)(1 - \eta)\mu_u$, $(c + \rho)\theta_1 = \lambda(\theta_1)(1 - \eta)\mu_e$ and $c\theta_2 = \lambda(\theta_2)(1 - \eta)\mu_e$ into expressions for μ_u and μ_e :

$$\mu_u = \frac{y - z + (\lambda(\theta_1) + \lambda(\theta_2))\eta\mu_e}{r + \delta + 2\lambda(\theta_0)\eta}$$
(6.43)

$$\mu_e = \frac{y - z + \delta \mu_u}{r + 2\delta + (\lambda(\theta_1) + \lambda(\theta_2))\eta}$$
(6.44)

Direct comparison of these equations with (4.33) and (4.34) reveals that the decentralized labour market equilibrium with family search, competitive wage setting in the public submarket and bargaining in the informal submarket is constrained efficient if $\beta = \eta$. If the Hosios condition $\beta = \eta$ is not satisfied, welfare in the benchmark equilibrium is reduced due to the presence of congestion externalities in the informal job market: unemployed workers searching for jobs reduce the probability for the remaining unemployed workers to find a job. Similarly new vacancies open in the informal job market reduce the probability for other firms to hire a worker. Overall, economic agents on the same side of the market create negative search externalities, while agents on different sides of the market create positive search externalities for each other. If the bargaining power parameter is low ($\beta < \eta$) welfare in the benchmark model with strong ties is reduced due to the fact that wages w_2 in the informal job market are too low, while the market tightness θ_2 is too high. In contrast, if the bargaining power is high ($\beta > \eta$) welfare is reduced as a result of wages w_2 being too high while the market tightness θ_2 is too low.

Welfare implications of the network externality in a model without income sharing can be analyzed on the basis of lemma 3, showing that a regime switch from full income sharing within a pair of connected workers to the economy without income sharing and under the simplifying assumption $r \to 0$, is equivalent to an exogenous shift in the bargaining power parameter from β to $\tilde{\beta}$. This means that at low values of β network externality has a reducing effect on the congestion externality through a positive effect on wages w_2 . Higher wages attract more firms to the informal job market implying a higher value of θ_2 . The overall welfare is then increased as a result of higher output produced in the informal job market. On the contrary, if the bargaining power parameter is high ($\beta > \eta$) network externality has an amplifying effect on the congestion externality. This means an additional upward distortion on wages w_2 corresponding to a lower market tightness θ_2 and a lower equilibrium welfare. These results are summarized in corollary 2:

Corollary 2: Let $\beta = \eta$, where $\eta = -(\partial q(\theta)/\partial \theta)(\theta/q(\theta))$ – elasticity of the job filling rate $q(\theta)$, then the decentralized competitive search equilibrium with bargaining in the informal job market and

- strong social ties described in proposition 1 is constrained efficient;
- weak social ties described in proposition 3 is inefficient. Job creation in the informal job market θ₂ and in the low wage segment of the public job market θ₀ are both insufficient, while θ₁ is excessive for r → 0.

Corollary 2 shows that welfare in the decentralized search equilibrium with weak ties does not achieve the maximum level at the Hosios value of the bargaining power $\beta = \eta$. The reason for this inefficiency is a positive wage bias in the informal job market, where the implied bargaining power of workers $\tilde{\beta}$ is larger than $\beta = \eta$. This means that the optimal actual bargaining power β^* in an economy with weak ties can be obtained by setting $\tilde{\beta} = \eta$, so that:

$$\frac{\eta - \beta^*}{1 - \beta^*} = \frac{\delta \eta (\lambda(\theta_1) + \lambda(\theta_2) - \lambda(\theta_0))}{2(\delta + \lambda(\theta_0)\eta)(\delta + \lambda(\theta_1) + \lambda(\theta_2))}$$
(6.45)

where variables θ_0 , θ_1 , θ_2 constitute an optimal choice of the social planner (for $r \to 0$) and can be obtained from proposition 3 by setting $\tilde{\beta} = \eta$ (see figure 5).



Figure 5: The optimal bargaining power in an economy with weak ties

The optimal bargaining power $\beta^* < \eta$ leads to $\tilde{\beta} = \eta$ and delivers constrained efficiency in a labour market model with weak ties.

7 Conclusions

This paper investigates the implications of job search through personal contacts on social welfare and wage dispersion in an equilibrium model with matching frictions. Upon entry firms have an option to post a high cost vacancy in the public job market or a low cost vacancy in the informal job market. Vacancy information in the informal submarket is only transmitted through employed personal contacts. In the benchmark model of the paper workers are grouped into a continuum of two-person households with a full income and information sharing between the members. Therefore this study combines the literature on joint job search with a focus on income sharing within a family and the literature on social networks with a focus on information sharing.

This paper shows that unemployed workers in mixed households gain from an additional option to screen jobs in the informal labour market, which is a result of information transmission from the employed to the unemployed household members. Ex-post differentiation of workers by social capital reflecting differences in the employment status of a connection gives rise to endogenous wage dispersion among equally productive risk-neutral workers. This is an extension of the result by Ek and Holmlund (2010) stating that risk-aversion is a necessary condition for wage dispersion in an equilibrium model with family job search. Moreover, the model exhibits a positive correlation in the employment status of household members observed in a number of empirical studies.

Wages in the public job market are set via the mechanism of competitive search utilizing the link between the probability to fill a vacancy and the posted wage offer. Endogenous heterogeneity of workers with respect to their reservation wages induces a segmentation in the public job market. Firms in a low wage segment of the public job market target at workers with low social capital and unemployed personal contacts, in contrast firms in a high wage segment target at workers with high social capital. Wages in the informal job market are set through individual bargaining. This highlights the non-competitive nature of wages paid in jobs obtained through personal contacts and allows for the possibility of wage penalties or wage premiums between the public and the informal job market.

Further, this paper proves that search equilibrium with competition in the public job market and bargaining in the informal market is unique and constrained efficient at the Hosios value of the bargaining power parameter. The new contribution of the paper is then to show that the efficient resource allocation is associated with a maximum wage dispersion in the public job market. This is due to the fact that the total output created in the informal job market is maximized at the efficient allocation, implying the highest value of the employed social contact granting access to the informal market and leading to the maximum heterogeneity in the reservation wages of workers.

The model is further extended to relax the assumption of income sharing, a pair of connected workers can then be interpreted as friends or acquaintances helping each other to find a job (weak ties). From the perspective of positive analysis this paper shows that weak ties yield higher wages in the informal submarket than family contacts, which is supported by the existing empirical evidence. Considered from the normative perspective, higher wages in jobs obtained through personal contacts lead to lower job creation in the informal job market, so the optimal bargaining power in an economy with weak ties is below the traditional Hosios value. If the bargaining power of workers is low, the network inefficiency is neutralizing the classical inefficiency from search frictions. This leads to an unambiguous increase in the total output. The effects on workers with different levels of social capital are however adverse. Unemployed workers with low social capital gain from a higher probability to find jobs in a low wage segment of the public job market despite a corresponding reduction in wages. On the contrary, unemployed workers with high social capital are confronted with a lower probability to find a job, but are compensated by higher wages. Furthermore welfare and output are reduced and the effects on workers with different level of social capital are reversed if the bargaining power parameter and wages in the informal job market are high.

8 Appendix

Appendix I: Proof of proposition 2:

Differentiate the job creation condition JC_0 with respect to β :

$$-\frac{(c+\rho)q'(\theta_0)}{q^2(\theta_0)}\frac{\partial\theta_0}{\partial\beta} = (1-\eta)\Big[\frac{\partial S^0}{\partial\theta_0}\frac{\partial\theta_0}{\partial\beta} + \frac{\partial S^0}{\partial\alpha(\theta_1)}\frac{\partial\alpha(\theta_1)}{\partial\beta}\Big]$$

Let $\eta_s^0 = -(\partial S^0/\partial \theta_0)(\theta_0/S^0)$ – elasticity of the total surplus S^0 with respect to θ_0 , $\eta_s^0 > 0$. This yields:

$$\frac{\partial \theta_0}{\partial \beta} (\eta + \eta_s^0) = \frac{\theta_0}{S^0} \frac{\partial S^0}{\partial \alpha(\theta_1)} \frac{\partial \alpha(\theta_1)}{\partial \beta}$$

This means that the sign of $\partial \theta_0 / \partial \beta$ is the same as the sign of $\partial \alpha(\theta_1) / \partial \beta$. Now differentiate the job creation condition JC_2 to obtain:

$$\eta \frac{\partial \theta_1}{\partial \beta} \frac{S^1}{\theta_1} = \frac{\partial S^1}{\partial \theta_0} \frac{\partial \theta_0}{\partial \beta} + \frac{\partial S^1}{\partial \alpha(\theta_1)} \frac{\partial \alpha(\theta_1)}{\partial \beta}$$

Insert expression for $\partial \theta_0 / \partial \beta$ and let $\eta_s^1 = -(\partial S^1 / \partial \theta_0)(\theta_0 / S^1)$ – elasticity of the total surplus S^1 with respect to θ_0 , $\eta_s^1 > 0$. This yields:

$$\eta \frac{\partial \theta_1}{\partial \beta} = \Big[\frac{\partial S^1}{\partial \alpha(\theta_1)} \frac{\theta_1}{S^1} - \frac{\eta_s^1}{\eta + \eta_s^0} \frac{\partial S^0}{\partial \alpha(\theta_1)} \frac{\theta_1}{S^0} \Big] \frac{\partial \alpha(\theta_1)}{\partial \beta}$$

Denote expression in the square bracket by Ω , so that:

$$\Omega \equiv \frac{\partial S^1}{\partial \alpha(\theta_1)} \frac{\theta_1}{S^1} - \frac{\eta_s^1}{\eta + \eta_s^0} \frac{\partial S^0}{\partial \alpha(\theta_1)} \frac{\theta_1}{S^0} < 0$$

As follows from lemma 1 Ω is negative, so the sign of $\partial \theta_1 / \partial \beta$ is opposite to the sign of $\partial \alpha(\theta_1) / \partial \beta$. Further differentiate equation $q(\theta_2) = c(1-\eta)q(\theta_1)/[(c+\rho)(1-\beta)]$ with respect to β to obtain:

$$\frac{\partial \theta_2}{\partial \beta} \frac{\beta}{\theta_2} = \frac{\partial \theta_1}{\partial \beta} \frac{\beta}{\theta_1} - \frac{\beta}{\eta(1-\beta)}$$

Therefore if $\partial \theta_1 / \partial \beta < 0$, it follows that $\partial \theta_2 / \partial \beta < 0$. Differentiate variable $\eta \alpha(\theta_1) = \eta \lambda(\theta_1) + \beta \lambda(\theta_2)$ with respect to β :

$$\eta \frac{\partial \alpha(\theta_1)}{\partial \beta} = \eta \lambda'(\theta_1) \frac{\partial \theta_1}{\partial \beta} + \beta \lambda'(\theta_2) \frac{\partial \theta_2}{\partial \beta} + \lambda(\theta_2)$$

This equations implies that if $\partial \theta_1 / \partial \beta > 0$ and therefore $\partial \alpha(\theta_1) / \partial \beta < 0$ then it is true that $\partial \theta_2 / \partial \beta < 0$. Insert expression for $\partial \theta_2 / \partial \beta$ to obtain:

$$\eta \frac{\partial \alpha(\theta_1)}{\partial \beta} = \frac{\partial \theta_1}{\partial \beta} \left[\eta \lambda'(\theta_1) + \beta \lambda'(\theta_2) \frac{\theta_2}{\theta_1} \right] + \lambda(\theta_2) \left[1 - \frac{\beta(1-\eta)}{\eta(1-\beta)} \right]$$

Finally, insert $\partial \alpha(\theta_1)/\partial \beta$ into the equation for $\partial \theta_1/\partial \beta$:

$$\frac{\partial \theta_1}{\partial \beta} \Big(\eta^2 - \Omega \Big[\eta \lambda'(\theta_1) + \beta \lambda'(\theta_2) \frac{\theta_2}{\theta_1} \Big] \Big) = \Omega \lambda(\theta_2) \frac{\eta - \beta}{\eta(1 - \beta)}$$

Since $\Omega < 0$ expression in the square bracket on the left-hand side is positive, therefore the sign of $\partial \theta_1 / \partial \beta$ is solely determined by the sign of $\Omega(\eta - \beta)$ on the right hand-side, which is negative if $\beta < \eta$ and it is positive if $\beta > \eta$. This means:

$$\frac{\partial \theta_1}{\partial \beta} < 0 \quad \frac{\partial \theta_2}{\partial \beta} < 0 \quad \frac{\partial \alpha(\theta_1)}{\partial \beta} > 0 \quad \frac{\partial \theta_0}{\partial \beta} > 0 \quad \text{if} \quad \beta < \eta$$
$$\frac{\partial \theta_1}{\partial \beta} > 0 \quad \frac{\partial \theta_2}{\partial \beta} < 0 \quad \frac{\partial \alpha(\theta_1)}{\partial \beta} < 0 \quad \frac{\partial \theta_0}{\partial \beta} < 0 \quad \text{if} \quad \beta > \eta$$

Appendix II: Proof of lemma 3. Worker surplus values R_e^1 and R_e^2 are given by:

$$(r+2\delta)R_e^1 = w_1 - rU_e + \delta(W_u^1 - W_u^0) + \delta(W_u^0 - U_e)$$

(r+2\delta)R_e^2 = w_2 - rU_e + \delta(W_u^2 - W_u^0) + \delta(W_u^0 - U_e)

which gives rise to the following derivative functions:

$$\frac{\partial R_e^1}{\partial w_1} = \frac{1}{r+\delta} + \frac{\delta}{r+2\delta} \frac{\partial W_u^0}{\partial w_1} \quad \text{and} \quad \frac{\partial R_e^1}{\partial w_1} = \frac{\delta}{r+2\delta} \frac{\partial W_u^0}{\partial w_1}$$

The worker indifference curve $U_e = cst$ is given by:

$$(r+\delta)U_e = z + \lambda(\theta_1)R_e^1 + \lambda(\theta_2)R_e^2 + \delta U$$

with a slope:

$$\lambda'(\theta_1)\frac{\partial\theta_1}{\partial w_1}R_e^1 + \lambda(\theta_1)\frac{\partial R_e^1}{\partial w_1} + \lambda(\theta_2)\frac{\partial R_e^2}{\partial w_1} + \delta\frac{\partial U}{\partial w_1} = 0$$

$$\lambda'(\theta_1)\frac{\partial \theta_1}{\partial w_1}R_e^1 + \frac{\lambda(\theta_1)}{r+\delta} + \delta \frac{\partial W_u^0}{\partial w_1} \Big[\frac{(\lambda(\theta_1) + \lambda(\theta_2))}{r+2\delta} + \frac{\lambda(\theta_0)}{r+2\lambda(\theta_0)}\Big] = 0$$

where the asset value W_u^0 can be obtained from:

$$(r+\delta)W_u^0 = w_0 + \delta U + (\lambda(\theta_1) + \lambda(\theta_2))\Delta\Phi$$

and $\Delta \Phi$ is given by:

$$(r+2\delta)\Delta\Phi = w_0 - rW_u^0 - \delta(W_u^0 - U_e)$$

The derivative of W_u^0 with respect to w_1 is then:

$$(r+\delta)\frac{\partial W_u^0}{\partial w_1} = \delta \frac{\partial U}{\partial w_1} + \lambda'(\theta_1)\frac{\partial \theta_1}{\partial w_1}\Delta\Phi - \frac{(\lambda(\theta_1) + \lambda(\theta_2))}{r+2\delta}\frac{\partial W_u^0}{\partial w_1}(r+\delta)$$

$$(r+\delta)\frac{\partial W_u^0}{\partial w_1} \Big[1 - \frac{\delta\lambda(\theta_0)}{(r+\delta)(r+2\lambda(\theta_0))} + \frac{(\lambda(\theta_1) + \lambda(\theta_2))}{r+2\delta} \Big] = \lambda'(\theta_1)\frac{\partial\theta_1}{\partial w_1}\Delta\Phi$$

giving rise to equation (5.37) in lemma 3.

Proof of proposition 3. Consider the case $r \to 0$, at the implied bargaining power $\tilde{\beta}$ surplus values S^0 and S^1 are given by:

$$S^{1} = \frac{(y-z)(\delta + \lambda(\theta_{0})\eta)}{\lambda(\theta_{0})\eta(2\delta + \eta\lambda(\theta_{1}) + \tilde{\beta}\lambda(\theta_{2})) + \delta^{2}} \qquad S^{0} = \frac{(y-z)(\delta + \eta\lambda(\theta_{1}) + \tilde{\beta}\lambda(\theta_{2}))}{\lambda(\theta_{0})\eta(2\delta + \eta\lambda(\theta_{1}) + \tilde{\beta}\lambda(\theta_{2})) + \delta^{2}}$$

The surplus difference ΔU is then:

$$\delta \Delta U + (\lambda(\theta_1) + \lambda(\theta_2)) \Delta \Phi = (\eta \lambda(\theta_1) + \tilde{\beta} \lambda(\theta_2)) S^1 - \lambda(\theta_0) \eta S^0$$

$$2\Delta\Phi(\delta+\lambda(\theta_1)+\lambda(\theta_2)) = \frac{(y-z)\delta(\eta\lambda(\theta_1)+\tilde{\beta}\lambda(\theta_2)-\eta\lambda(\theta_0))}{\lambda(\theta_0)\eta(2\delta+\eta\lambda(\theta_1)+\tilde{\beta}\lambda(\theta_2))+\delta^2}$$

The implied bargaining power $\tilde{\beta}$ is then given by:

$$\tilde{\beta} = \beta + (1-\beta)\frac{\Delta\Phi}{S^2} = \beta + (1-\beta)\frac{\delta(\eta\lambda(\theta_1) + \beta\lambda(\theta_2) - \eta\lambda(\theta_0))}{2(\delta + \eta\lambda(\theta_0))(\delta + \lambda(\theta_1) + \lambda(\theta_2))}$$

Extracting $\tilde{\beta}$ from the above equation produces expression (5.39) in proposition 3.

Appendix III: Proof of proposition 4:

The current value Hamiltonian for the social planner problem is:

$$H = p_u 2z + p_m (z + y) + p_e 2y - (c + \rho)\theta_0 2p_u - (c + \rho)\theta_1 p_m - c\theta_2 p_m$$

+ $\mu_u [2\lambda(\theta_0)p_u - \delta p_m] + \mu_e [(\lambda(\theta_1) + \lambda(\theta_2))p_m - 2\delta p_e]$
+ $\mu [0.5 - [p_u + p_m + p_e]]$

where μ_u , and μ_e are costate variables corresponding to p_u and p_e respectively. The optimal social planner solution must satisfy:

$$\begin{array}{lll} \displaystyle \frac{\partial H}{\partial p_m} &=& z+y-(c+\rho)\theta_1-c\theta_2-\mu_u\delta+\mu_e(\lambda(\theta_1)+\lambda(\theta_2))-\mu=0\\ \displaystyle \frac{\partial H}{\partial p_u} &=& 2z-2(c+\rho)\theta_0+\mu_u2\lambda(\theta_0)-\mu=-r\mu_u\\ \displaystyle \frac{\partial H}{\partial p_e} &=& 2y-\mu_e2\delta-\mu=r\mu_e\\ \displaystyle \frac{\partial H}{\partial \theta_0} &=& -(c+\rho)2p_u+\mu_u2\lambda'(\theta_0)p_u=0\\ \displaystyle \frac{\partial H}{\partial \theta_1} &=& -(c+\rho)p_m+\mu_e\lambda'(\theta_1)p_m=0\\ \displaystyle \frac{\partial H}{\partial \theta_2} &=& -cp_m+\mu_e\lambda'(\theta_2)p_m=0 \end{array}$$

9 References

ADDISON J.T., P. PORTUGAL (2002): "Job Search Methods and Outcomes", Oxford Economic Papers, 54(3): 505-533.

BENTOLILA S., C. MICHELACCI, SUAREZ J. (2010): "Social Contacts and Occupational Choice", *Economica*, 77(305): 20-45.

BEWLEY T. (1999): "Why wages don't fall during a recession", Harvard University Press.

BLAZQUEZ M., M. JANSEN (2008): "Search, Mismatch and Unemployment", *European Economic Review*, 52: 498-526.

CAHUC P, F. FONTAINE (2009): "On the Efficiency of Job Search with Social Networks", *Journal of Public Economic Theory*, 11(3): 411-439.

CALVO-ARMENGOL A., M.O. JACKSON (2004): "The Effects of Social Networks on Employment and Inequality", *The American Economic Review*, 94(3): 426-454.

CALVO-ARMENGOL A., M.O. JACKSON (2007): "Networks in Labor Markets: Wage and Employment Dynamics and Inequality", *Journal of Economic Theory*, 132: 27-46.

CAPELLARI L., K. TATSIRAMOS (2010): "Friends Networks and Job Finding Rates", *IZA Discussion paper* No. 5240.

COLEMAN J.S. (1988): "Social capital and the Creation of Human capital", *The American Journal of Sociology*, Supplement: Organizations and Institutions: Sociological and Economic Approaches to the Analysis of Social Structure, 94: 95-120.

DELATTRE E., M. SABATIER (2007): "Social Capital and Wages: an Econometric Evaluation of Social Networking's Effects", *Labour*, 21(2): 209-236

DIAMOND P.A. (1982): "Wage Determination and Efficiency in Search Equilibrium", *Review of Economic Studies* 49: 217-227.

EK S., B. HOLMLUND (2010): "Family Job Search, Wage Bargaining, and Optimal Unemployment Insurance", *The B.E. Journal of Economic Analysis and Policy*, 10(1): 47.

FONTAINE F. (2004): "Do Workers Really Benefit from Their Social Networks?", *IZA Discussion paper* No. 1282.

FONTAINE F. (2007): "A Simple Matching Model with Social Networks", *Economics Letters*, 94(3): 396-401.

FONTAINE F. (2008): "Why Are Similar Workers Paid Differently? The Role of Social Networks", *Journal of Economic Dynamics and Control*, 32(12): 3960-3977.

GAUTIER P. (2002): "Unemployment and Search Externalities in a Model with Heterogeneous Jobs and Workers", *Economica*, 69: 21-40.

GRANOVETTER M. (1973): "The Strength of Weak Ties", American Journal of Sociology, 78(6): 1360-1380.

GRANOVETTER M. (1995): "Getting a Job: A Study of Contacts and Careers", Chicago: University of Chicago Press.

GULER B., F. GUVENEN, G.L. VIOLANTE (2009): "Joint-Search Theory: New Opportunities and New Frictions", *NBER Working Paper* No. 15011.

HOSIOS A.J. (1990): "On the Efficiency of Matching and Related Models of Search and Unemployment", *Review of Economic Studies*, 57: 279-298.

IOANNIDES Y.M., L. DATCHER LOURY (2004): "Job Information Networks, Neighborhood Effects, and Inequality", *Journal of Economic Literature*, 42(4): 1056-1093.

KRAMARZ F., O. NORDSTROM SKANS (2010): "With a Little Help from my ... Parents? Family Networks and Youth Labour Market Entry", *Working paper*. MARGOLIS D.M., V. SIMMONET (2003): "Educational Track, Networks and Labor Market Outcomes", *IZA Discussion paper*, No. 699.

MOEN E.R. (1997): "Competitive Search Equilibrium", *Journal of Political Economy*, 105(2): 385-411.

MONTGOMERY J.D. (1991): "Social Networks and Labor-Market Outcomes: Toward an Economic Analysis", *The American Economic Review*, 81(5): 1408-1418.

MONTGOMERY J.D. (1992): "Job Search and Network Composition: Implications of the Strength-Of-Weak-Ties Hypothesis", *American Sociological Review*, 57(5): 586-596.

MONTGOMERY J.D. (1994): "Weak Ties, Employment, and Inequality: An Equilibrium Analysis", *The American Journal of Sociology*, 99(5): 1212-1236.

MORTENSEN D. (1982): "Property Rights and Efficiency in Mating, Racing, and Related Games", *American Economic Review*, 72: 968-979.

MORTENSEN D.T., C.A. PISSARIDES (1994): "Job Creation and Job Destruction in the theory of unemployment", *Review of Economic Studies*, 61: 397-415.

MORTENSEN D.T., T. VISHWANATH (1994): "Personal Contacts and Earnings It is who you know!", *Labour Economics*, 1: 187-201.

PELLIZZARI M. (2010): "DO FRIENDS AND RELATIVES REALLY HELP IN GETTING A GOOD JOB?", *Industrial and Labor Relations Review*, 63(3): 494-510.

PISSARIDES, C.A. (1984): "Search Intensity, Job Advertising, and Efficiency", *Journal of Labor Economics*, 2: 128-143.

PISSARIDES C.A. (1985): "Short-run Equilibrium Dynamics of Unemployment, Vacancies and Real Wages", *American Economic Review*, 75: 676-90.

PISSARIDES C. A. (2000): Equilibrium Unemployment Theory. The MIT Press. Cambridge.

PISTAFERRI L. (1999): "Informal Networks in the Italian Labor Market", *Giornale degli Economisti e Annali di Economia*", 58(3-4): 355375.

PONZO M., V. SCOPPA (2010): "The Use of Informal Networks in Italy: Efficiency or Favoritism?", *Journal of Socio-Economics*, 39(1): 89-99

STAIGER D. (1990): "The Effect of Connections on the Wages and Mobility of Young Workers", *Dissertation*, MIT.

SYLOS LABINI M. (2004): "Social Networks and Wages. It is all about connections!", *LEM paper series* 2004/10, Pisa, Italy.